Author's response to reviews

Title: Particulate air pollution and chronic ischemic heart disease in the eastern United States: a county level ecological study using satellite aerosol data

Authors:

Zhiyong Hu (zhu@uwf.edu)
K Ranga Rao (rrao@uwf.edu)

Version: 3 Date: 8 June 2009

Author's response to reviews:

1. Tables are now in closed cell format using Microsoft Word Tables tool. Tables were divided into rows and columns and be oriented in a portrait page layout. Vertical lines are not shown in the tables.

2. References 13 and 39 have been revised with citation details.

3. A few typographical errors have been corrected.